Author’s response to reviews

Title: Healthy-Lifestyle Behaviors Associated with Overweight and Obesity in US Rural Children

Authors:

Alison Tovar (alison.tovar@tufts.edu)
Kenneth Chui (Kenneth.Chui@tufts.edu)
Vivica Kraak (vivica.kraak@gmail.com)
Julia Kuder (Julia.kuder@tufts.edu)
Raymond R Hyatt (Raymond.hyatt@tufts.edu)
Silvina Choumenkovitch (silvina.choumenkovitch@tufts.edu)
Alia Hastings (Alia.Hastings@tufts.edu)
Julia Bloom (Julia.Bloom@tufts.edu)
Christina D Economos (christina.economos@tufts.edu)

Version: 6 Date: 31 May 2012

Author’s response to reviews:

Given that the reviewer had no changes in the manuscript, we are re-submitting the manuscript as is.